Author’s response to reviews

Title: An integrative review of the methodology and findings regarding dietary adherence in end stage kidney disease

Authors:
Kelly Lambert (klambert@uow.edu.au)
Judy Mullan (jmullan@uow.edu.au)
Kylie Mansfield (kylie@uow.edu.au)

Version: 2  Date: 25 Sep 2017

Author’s response to reviews:

Dear Editor

Thank you for the opportunity to make minor revisions to our manuscript.

A declarations section has been uploaded. This has also been included below for your reference.

Declarations

- Ethics approval and consent to participate – not applicable

- Consent to publish - not applicable

- Availability of data and materials - All data generated or analysed during this study are included in this published article.
- Competing interests – nil

- Funding: This work was has been conducted with financial support from the Australian Government Research Training Program Scholarship and the King and Amy O’Malley Trust

- Authors' Contributions: Kelly Lambert: Conceptualisation of study design, data collection, data analysis, primary responsibility for writing the article. Judy Mullan: Refinements to study design, data analysis, writing the article. Kylie Mansfield: Refinements to study design, data analysis, writing the article.

- Acknowledgements - not applicable